Author’s response to reviews

Title: Antibiotic resistance patterns of bacteria causing urinary tract infections in the elderly living in nursing homes versus the elderly living at home: an observational study

Authors:

Mark Fagan (marfaga@online.no)
Morten Lindbæk (morten.lindbak@medisin.uio.no)
Nils Grude (nigrud@siv.no)
Harald Reiso (harald.reiso@medisin.uio.no)
Maria Romøren (maria.romoren@medisin.uio.no)
Dagfinn Skare (dagfinn.skaare@siv.no)
Dag Berild (dag.berild@medisin.uio.no)

Version: 3 Date: 25 May 2015

Author’s response to reviews:

Thank you again for evaluating our manuscript. In your last correspondence you asked: "Please include the Authors and their contributions in the Authors’ Contribution section."

I have rewritten this section specifying with initials the specific contributions of the authors. The following is a copy of the text in the manuscript:

All authors (MF, ML, NG, HR, MR, DS and DB) participated in the design of the study. ML, MR and DS collected the data. MF analyzed the data and drafted the manuscript. MF, ML, NG, HR, MR, DS and DB contributed to the interpretation of the analyses, critical reviews and revisions, and final approval of the paper.